Author's response to reviews

Title: Influence network linkages across treatment conditions in a randomized controlled trial of two implementation strategies for scaling up evidence-based practices in public youth-serving systems

Authors:

Lawrence A Palinkas (palinkas@usc.edu)
Ian W Holloway (holloway@publicaffairs.ucla.edu)
Eric Rice (ericr@usc.edu)
C. Hendricks Brown (hendricks.brown@northwestern.edu)
Thomas W Valente (tvalente@usc.edu)
Patricia A Chamberlain (pattic@oslc.org)

Version: 5 Date: 12 October 2013

Author's response to reviews:

Dear Editors: As requested, we are uploading a clean version of the copyedited version of our manuscript. We accept all of the changes made by the copyeditor with one exception. Dr. Brown's affiliating has been changed to reflect his current position. He is now with the Department of Psychiatry and Behavioral Sciences at the Feinberg School of Medicine at Northwestern University. Thank you once again for accepting our manuscript for publication.